Author's response to reviews

Title: Using theatre in education in a traditional lecture oriented medical curriculum

Authors:

Pemra C Ünal (pcunalan@gmail.com)
Arzu Uzuner (arzuuzuner@gmail.com)
Serap Çifçili (serapcifcili@gmail.com)
Mehmet Akman (makman4@gmail.com)
Sertaç Hancıoğlu (dr_sertach@hotmail.com)
Hans O Thulesius (hans.thulesius@ltkronoberg.se)

Version: 6 Date: 30 October 2009

Author's response to reviews:

In this revision-3 only the affiliation of the 5th author was changed and updated.
With my best regards,
Pemra C.ÜNALAN, MD
Corresponding author